Author's response to reviews

Title: Gender differences in risk for intimate partner violence in South Korea

Authors:

Minjee Lee (minjee-lee@uiowa.edu)
Katherine M Stefani (kt.stefani@gmail.com)
Eun-Cheol Park (ecpark@yuhs.ac)

Version: 3
Date: 7 November 2013

Author's response to reviews: see over
November 6, 2013

Editor-in-Chief
BMC Public Health

RE: Manuscript No. 1877908484112474

Dear Editor:

We are so honored to get our manuscript entitled “Gender Differences in Risk for Intimate Partner Violence in South Korea.” considered by the Journal. We also would like to thank you so much for the opportunity to do formatting corrections. As suggested, we removed the 'Disclosure of Competing Interest', 'Acknowledgments' and 'Authors' contributions' sections from the title page, and placed after the Conclusions/Abbreviations.

We believe that the changes made in response to the comments have significantly strengthened the manuscript, and thank you.

I hope you find the draft acceptable and look forward to hearing from you.

Thank you for your time and consideration.

Sincerely,

Eun-Cheol Park

Eun-Cheol Park, MD, Ph.D.
Department of Preventive Medicine and Public Health
Institute of Health Services Research
Yonsei University College of Medicine
Seoul, Korea
Tel: 82-2-2228-1862
Fax: 82-2-392-8133
Email: ecpark@yuhs.ac